Author's response to reviews

Title: Comparison of measurement methods with a mixed effects procedure accounting for replicated evaluations (COM3PARE): Method comparison algorithm implementation for head and neck IGRT positional verification

Authors:

Anuradha Roy (Anuradha.Roy@utsa.edu)
Clifton D Fuller (cdfuller@mdanderson.org)
David I Rosenthal (dirosenthal@mdanderson.org)
Charles R Thomas Jr (thomasch@ohsu.edu)

Version: 2 Date: 19 July 2014

Author's response to reviews:

Corrections made as per pre-review specifications.